\documentclass[11pt,twoside]{article}
\usepackage{url}
\input{macros-course}

\newcommand{\comment}[1]{\textsf{[[[#1]]]}}
\newcommand{\computertype}[1]{\texttt{{\small #1}}}
\addtolength{\textheight}{1in}
\begin{document}

\handout{topics}{Exam Topics}{May 20, 2001}

This handout lists the topics that will be covered on the final exam.
\begin{enumerate}
\item Proofs
\begin{itemize}
\item Basic Induction
\item Strong Induction
\item Structural Induction
\item Recursive Programs and Induction
\end{itemize}
\item Sets
\item Relations
\begin{itemize}
\item Equivalence Relations
\item Partial Orders
\end{itemize}
\item Graphs, Trees, Tours and Paths
\item Computational Processes
\begin{itemize}
\item Invariants
\item Termination
\item Step Bounds
\end{itemize}
\item Series, Summations and Products
\begin{itemize}
\item Annuities
\end{itemize}
\item Estimation and Asymptotics
\item Recurrences
\item Cardinality
\item Functions and Bijections
\item Pigeonhole Principle
\item Counting
\item Infinite Sets
\item Probability
\begin{itemize}
\item Conditional Probability
\item Set Theory and Probability
\end{itemize}
\item Random Variables
\newpage
\item Expected Value
\begin{itemize}
\item Expected Value of a Product
\item Deviation from the Mean
\item Probability of k Events
\end{itemize}
\item Random Walks
\end{enumerate}
\end{document}
